Author's response to reviews

Title: Availability and structure of primary medical care services and population health and health care indicators in England

Authors:

- Martin C Gulliford (martin.gulliford@kcl.ac.uk)
- Ruth H Jack (ruth.jack@kcl.ac.uk)
- Geoffrey Adams (gpadams4@hotmail.com)
- Dr Obioha C Ukoumunne (obioha.ukoumunne@kcl.ac.uk)

Version: 4 Date: 11 Jun 2004

Re-formatted paper